Author's response to reviews

Title: Gastric transit and small intestinal transit time and motility assessed by a magnet tracking system

Authors:

Jonas Worsoe (jonas.worsoe@gmail.com)
Lotte Fynne (lfynne@hotmail.com)
Tine Gregersen (tine_gregersen@hotmail.com)
Vincent Schlageter (vincent.schlageter@motilis.com)
Lisbet A. Christensen (lisbchte@rm.dk)
Jens F. Dahlerup (jensdahl@rm.dk)
Nico J.M. Rijkhoff (nr@hst.aau.dk)
Søren Laurberg (SOERLAUR@rm.dk)
Klaus Krogh (klaukrog@rm.dk)

Version: 6 Date: 17 December 2011

Author's response to reviews:

To the Production team

We have made the changes as suggested.

One problem: The abstract in the submission system and in the manuscript should be identical; but in the submission system the abstract cannot contain uppercase letters or Greek letters. We would like that to appear in the final manuscript. What can we do?

Kind Regards Jonas Worsoe